Author’s response to reviews

Title: Emergency Management in Health: Key Issues and Challenges in the UK

Authors:

Andrew CK Lee (andrew.lee@sheffield.ac.uk)
Wendy Phillips (wendy.phillips@hpa.org.uk)
Kirsty Challen (kirstychallen@hotmail.com)
Steve Goodacre (s.goodacre@sheffield.ac.uk)

Version: 2 Date: 13 June 2012

Author’s response to reviews:

Addition of details of the organisation providing ethical approval for the study have been made in the methods section as instructed.